Author's response to reviews

Title: Osteogenic Differentiation of Dental Pulp Stem Cells under the Influence of Three Different Materials

Authors:

Sumaiah A Ajlan (dr-ajlan2006@hotmail.com)
Nahid Y Ashri (nashri00@yahoo.com)
Abdullah M Aldahmash (a.aldahmash@gmail.com)
May S Alnbaheen (maysalemalnbaheen@gmail.com)

Version: 3 Date: 26 September 2015

Author's response to reviews: see over
Dear Chief Editor.

Submitted is the reviewed version of the manuscript entitled “Osteogenic Differentiation of Dental Pulp Stem Cells under the Influence of Three Different Materials”, after doing the requested adjustments, which include:

- Moving the funding details into the acknowledgment section
- Removing figure legends from the figure documents
- Placing tables at the end of the main document

I would like to thank you as well as your reviewer board for taking the time considering this manuscript, hoping that the review process finds it acceptable for publication in your esteemed journal.

Sincerely yours,

Sumaiah A. Ajlan
Lecturer
Department of Periodontics and Community Dentistry
King Saud University
Riyadh, Saudi Arabia